Author's response to reviews

Title: Glioma stem cells are more aggressive in the recurrent tumor than in the primary tumor and they both can be maintained for long-terms in vitro

Authors:

Huang Qiang (hq1936@yahoo.com.cn)
Zhang Quan-Bin (quanbinzhang@yahoo.com.cn)
Dong Jun (didongiun@163.com)
Wu Yin-Yan (wchywy@yahoo.com.cn)
Shen Yun-Tian (shenyuntian2001@yahoo.com.cn)
Zhao Yao-Dong (dm0920@gmail.com)
Zhu Yu-De (zyddzy2000@yahoo.com.cn)
Diao Yi (dpdy118@yahoo.com.cn)
Wang Ai-Dong (eagle.wangad@yahoo.com.cn)
Lan Qing (szlqz@pub.sz.jsinfo.net)

Version: 2 Date: 23 April 2008

Author's response to reviews:

Dear Editors,

My revisions is mainly focused on the title of this manuscript, I have changed a few words in the title to make the title more compatible with the manuscript.

Thank you for your kind consideration!

Yours sincerely,

Jun Dong